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Fill summary

Duration of stable beams up to almost 12 hours.
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Last runs with TRD triggers since 22/04

285222 36.3 Mevents, trigger rejection HJT 3873 HNU 1250 HQU 433
285200 33.5 Mevents, trigger rejection HJT 3957 HNU 1098 HQU 353 (Benjamin)
280066 10. Mevents, trigger rejection HJT 3973 HNU 1374 HQU 479 (10/10/17)
In general agreement with previous year. More cross-checks for final confirmation?
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DQM plots

Almost all channels with HV (Yvonne) and FEDs off are labeled.
=⇒ Quite reasonable performance...
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Updated LME DQM plots

A few more channels will be labeled with HV and FED will be switched off.
Cross-checks of Yvonne.
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Updated connection for alidcswie030

after access on 18/04. Broken outlet/connection in O26 3618/08 confirmed, new
outlet O32 4218/02 (Alexander,Ole):

Firmware was updated for all spared PSUs on 20/04 (Alexander,Mateusz).
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To do

• Confirm the activation of TRD-based triggers: HJT, HNU, HQU.

• Finish disabling of FEDs and its list for AMORE according to the last runs.
But without TRD interventions on the DCS during the ongoing handshake.

• Tune the layout of the DQM LME plots (Ruediger).

• Take pedestal run.

• Check TRD configuration for negative polarity.

• Increase alarm thresholds for alidcswie30 channel0 (SEC06L01) from 136A to
140A due to alerts at 139A (Lucas)?

• +?
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